Author’s response to reviews

Title: Estrogen receptor (ER) signaling regulates the expression of the breast tumor kinase (BRK) in breast cancer cells

Authors:

Sayem Miah (smi@stowers.org)
Edward Bagu (edward.bagu@usask.ca)
Raghuveera Goel (raghu.goel@usask.ca)
Yetunde Ogunbolude (yetunde.ogunbolude@usask.ca)
Chenlu Dai (chenlu.dai@usask.ca)
Alison Ward (alison.ward@usask.ca)
Frederick Vizeacoumar (frederick.vizeacoumar@usask.ca)
Gerald Davies (gerald.davies@usask.ca)
Franco Vizeacoumar (franco.vizeacoumar@usask.ca)
Deborah Anderson (deborah.anderson@saskcancer.ca)
Kiven Lukong (kiven.lukong@usask.ca)

Version: 1 Date: 16 Jul 2018

Author’s response to reviews:

Technical Comments:

** Title page: Please check the order of the authors in the metadata (appears on the 1st page of the pdf), as it is different from the one listed in the manuscript text. To change the order of authors, please use drag and drop option.

** Abstract: The abstract must only include the following separate sections:

- Background: the context and purpose of the study
- Methods: how the study was performed and statistical tests used
- Results: the main findings
- Conclusions: brief summary and potential implications

Please revise your abstract structure to conform to the journal's submission guidelines.

** Consent for publication: We note that you have included a ‘Consent for publication’ section in the Declarations.

The 'Consent for publication' statement covers manuscripts that contain any individual person’s data in any form (including individual details, images or videos), for which consent to publish must be obtained from that person, or in the case of children, their parent or legal guardian.

If your manuscript does not contain any individual person’s data, please state “Not applicable” in this section.

** Funding: Please state clearly the role the funder(s) had in your study in the "funding" section of the declarations.

** Abbreviations: Please list all abbreviations used in your manuscript under the heading "Abbreviations" after the conclusions section. If no abbreviations are used in the manuscript, please state "Not applicable” in this section.

Editor's comments:

Please remove the ages and gender from supplementary tables 3 & 4 and replace them with age ranges but not gender. This will enable anonymity to be maintained.